Page 1

IAP, January 2000

CBCl MIT

Center for Biological and Computational Learning + AI Lab

MikeTalk : A Photorealistic Talking Facial Display

Tony Ezzat Tomaso Poggio

IAP, January 2000

CBCl MIT

TTVS Overview

� Input: Text � Output:

photorealistic talking face uttering text

Page 2

IAP, January 2000

CBCl MIT

TTVS Modules

IAP, January 2000

CBCl MIT

Phoneme Corpus

Page 3

IAP, January 2000

CBCl MIT

6 Consonantal Visemes

� Step 2

� extract one image per phoneme: viseme � group visemes together by visual similarity

IAP, January 2000

CBCl MIT

9 Vocalic Visemes (+ 1 SilenceViseme)

Page 4

IAP, January 2000

CBCl MIT

Optical Flow

� Forward warping A to B

� Forward warping B to A

� Blending

� Holefilling

� Load the correct viseme transitions

� Concatenate viseme transitions

� Sample the viseme transitions using audio durations

Page 6

IAP, January 2000

CBCl MIT

Further Work

� Learning of dynamics/coarticulation for smoother mouth motion � Visual Prosody: eye movements, eye brows, head movements � Viewpoint changes

IAP, January 2000

CBCl MIT

Applications

� Special Effects/Digital Actors � Desktop agents � Automated news/web page presenters